Author's response to reviews

Title: Acculturation and Obesity among Migrant Populations in High Income Countries - a Systematic Review

Authors:

Maryam Delavari (mdela@deakin.edu.au)
Anders Larrabee Sonderlund (als224@exeter.ac.uk)
Boyd Swinburn (boyd.swinburn@auckland.ac.nz)
David Mellor (david.mellor@deakin.edu.au)
Andre MN Renzaho (andre.renzaho@monash.edu)

Version: 2 Date: 21 November 2012

Author's response to reviews: see over
November 22, 2012

Dear BioMed Central Editorial Team

Re: MS: 3137643208525236 - Acculturation and Obesity among Migrant Populations in High Income Countries - a Systematic Review

Thank you for your recent email. Please find a version of the manuscript containing the changes submitted online as well as the cover letter below with the point-by-point description of the changes made.

- Type of article changed to research.
- Subject areas relevant changed to Women’s health and Public Health.
- A Background as well as Methods section heading is included in the Abstract.
- A 'Competing interests' is included in the manuscript. It is a section between the Conclusions and ‘Authors' contributions’.
- The ‘Authors' contribution’ is included in the manuscript after ‘competing interests’.
- Authors' information is included in the manuscript after ‘Authors' contributions’ section.

Thank you for your time and we look forward to hearing from you.

Kind regards,

Dr. Maryam Delavari